Author’s response to reviews

Title: Factors affecting acceptability of an email-based intervention to increase fruit and vegetable consumption.

Authors:

Emily J Kothe (emily.kothe@deakin.edu.au)
Barbara A Mullan (barbara.mullan@curtin.edu.au)

Version: 3
Date: 30 May 2014

Author’s response to reviews: see over
15 May 2014

Re: Kothe & Mullan. **Factors affecting acceptability of an email-based intervention to increase fruit and vegetable consumption**

MS: 7782683151310109

To Whom It May Concern,

Please find attached an amended version of the manuscript entitled ‘Factors affecting acceptability of an email-based intervention to increase fruit and vegetable consumption’

The changes made include:

1) The inclusion of line numbers
2) A Consent Statement added at line 72 & 73.
3) A competing interests statement at line 176.
4) An explanation of authors’ contributions at lines 179-183.

Kind regards,

Emily Kothe